Author's response to reviews

Title: Serum Protein Gamma-Glutamyl Hydrolase, Ig Gamma-3 Chain C Region, and Haptoglobin are Associated with the Syndromes of Pulmonary Tuberculosis in Traditional Chinese Medicine

Authors:

Ting-Ting Jiang (11318038@zju.edu.cn)
Chong Wang (knight_880102@163.com)
Li-Liang Wei (weililiang1981@163.com)
Xiao-Mei Yu (yuxiaomei_1@hotmail.com)
Li-Ying Shi (shi-lee-ying@163.com)
Dan-Dan Xu (dandanxu1220@163.com)
Zhong-Liang Chen (zlchen_m@163.com)
Ze-Peng Ping (pingzepeng@126.com)
Ji-Cheng Li (lijichen@zju.edu.cn)

Version: 4 Date: 19 May 2015

Author's response to reviews: see over
Dear Editor:

We have revised the manuscript in light of the editor’s comments. Here below is our description on revision according to the editor’s comments:

Section Editor:

Comment 1: I have read through the further revised manuscript and I found the revisions are appropriate. I would accept it for publication if authors can add one recent reference (see below) in their introduction somewhere between line 91 and 99, which is important to support their justification of doing this work. (Wang M, Guan X, Chi Y, Robinson N, Liu JP. Chinese herbal medicine as adjuvant treatment to chemotherapy for multidrug-resistant tuberculosis (MDR-TB): a systematic review of randomised clinical trials. Tuberculosis DOI: http://dx.doi.org/10.1016/j.tube.2015.03.003.)

Response 1: The reference was added in the introduction (line 92 and 99) in light of the editor’s comments, and the subsequent reference numbers was also revised successively. All revisions were marked in highlight.

A list of changes:

1. Page 5, line 92. Insert reference number [6].
3. Added references: 6, revised reference numbers 7-52.

Sincerely Yours,

Ji-Cheng Li, M.D.
Professor, Director
Institute of Cell Biology, Zhejiang University
Hangzhou, 310058, China
Fax & Tel: +86-571-88208088
E-mail: lijichen@zju.edu.cn